Author's response to reviews

Title: QTc interval prolongation in HIV-infected patients: a case-control study by 24-hour Holter ECG recording

Authors:

Alessandra Fiorentini (alessandrafiorentini17@gmail.com)
Nicola Petrosillo (nicola.petrosillo@inmi.it)
Angelo Di Stefano (angelo.distefano@uniroma1.it)
Stefania Cicalini (stefania.cicalini@inmi.it)
Laura Borgognoni (laura.borgognoni@uniroma1.it)
Evangelo Boumis (evangelo.boumis@inmi.it)
Luigi Tubani (tubani@libero.it)
Pierangelo Chinello (pierangelo.chinello@inmi.it)

Version: 2 Date: 10 August 2012

Author's response to reviews: see over
To the Editor of
BMC Cardiovascular Disorders

Sir,

enclosed please find the revised manuscript of our work entitled “QTc interval prolongation in HIV-infected patients: a case-control study by 24-hour Holter ECG recording”.

In the METHODS section, the requested ethics statement has been added as follows:

“Informed consent was obtained from all participants; all the investigations were performed in accordance with the principles of the Declaration of Helsinki. The research was approved by the ethics committee of our Institution”.

Authors: Alessandra Fiorentini Nicola Petrosillo Angelo Di Stefano
Stefania Cicalini Laura Borgognoni Evangelo Boumis
Luigi Tubani Pierangelo Chinello

The corresponding author is Pierangelo Chinello, MD
Second Infectious Diseases Division, “L. Spallanzani” National Institute for Infectious Diseases
via Portuense 292, 00149 Rome (Italy)
Tel.: 0039-06-55170294 Fax: 0039-06-55170315
e-mail: pierangelo.chinello@inmi.it

Kind Regards.
Rome, August 10th, 2012

Pierangelo Chinello, MD